

# eCommons@AKU

# Department of Pathology and Laboratory Medicine

Medical College, Pakistan

3-23-2020

# Comparison of clinical presentation and out-comes of chikungunya and dengue virus infections in patients with acute undifferentiated febrile illness from the Sindh region of Pakistan

Uzma Shahid Aga Khan University

Joveria Farooqi Aga Khan University, joveria.farooqi@aku.edu

Kelli L. Barr Baylor University, Waco, Texas, United States of America

Syed Faisal Mahmood *Aga Khan University*, faisal.mahmood@aku.edu

Bushra Jamil Aga Khan University, bushra.jamil@aku.edu

See next page for additional authors

Follow this and additional works at: https://ecommons.aku.edu/pakistan\_fhs\_mc\_pathol\_microbiol

Part of the Infectious Disease Commons, and the Pathology Commons

# **Recommended Citation**

Shahid, U., Farooqi, J., Barr, K. L., Mahmood, S., Jamil, B., Imtiaz, K., Azizullah, Z., Malik, F., Prakoso, D., Long, M. T., Khan, E. (2020). Comparison of clinical presentation and out-comes of chikungunya and dengue virus infections in patients with acute undifferentiated febrile illness from the Sindh region of Pakistan. *PLOS Neglected Tropical Diseases*, *14*(3), e0008086.

Available at: https://ecommons.aku.edu/pakistan\_fhs\_mc\_pathol\_microbiol/1190

# Authors

Uzma Shahid, Joveria Farooqi, Kelli L. Barr, Syed Faisal Mahmood, Bushra Jamil, Kehkashan Imtiaz, Zahida Azizullah, Faisal Malik, Dhani Prakoso, Maureen T. Long, and E. Khan



# G OPEN ACCESS

**Citation:** Shahid U, Farooqi JQ, Barr KL, Mahmood SF, Jamil B, Imitaz K, et al. (2020) Comparison of clinical presentation and out-comes of Chikungunya and Dengue virus infections in patients with acute undifferentiated febrile illness from the Sindh region of Pakistan. PLoS Negl Trop Dis 14(3): e0008086. https://doi.org/10.1371/journal.pntd.0008086

Editor: Marilia Sá Carvalho, Fundacao Oswaldo Cruz, BRAZIL

Received: July 25, 2019

Accepted: January 24, 2020

Published: March 23, 2020

**Copyright:** © 2020 Shahid et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data underlying the findings of this research are fully described and available without restriction at the time of publication in the manuscript as text, tables, and supporting information.

**Funding:** This work was funded by DEFENSE THREAT REDUCTION AGENCY (<u>www.drta.mil</u>/) HDTRA1-09-14-FRCWMD-BAA to ML. The funders had no role in study design, data collection and RESEARCH ARTICLE

Comparison of clinical presentation and out-comes of Chikungunya and Dengue virus infections in patients with acute undifferentiated febrile illness from the Sindh region of Pakistan

Uzma Shahid<sup>1</sup>, Joveria Q. Farooqi<sup>1</sup>, Kelli L. Barr<sup>2</sup>, S. Faisal Mahmood<sup>3</sup>, Bushra Jamil<sup>3</sup>, Kehkashan Imitaz<sup>1</sup>, Zahida Azizullah<sup>1</sup>, Faisal R. Malik<sup>1</sup>, Dhani Prakoso<sup>4</sup>, Maureen T. Long<sup>4</sup>, Erum Khan<sup>1</sup>\*

1 Department of Pathology and Laboratory Medicine, Aga Khan University, Karachi, Pakistan, 2 Department of Biology, Baylor University, Waco, Texas, United States of America, 3 Department of Medicine, Aga Khan University, Karachi, Pakistan, 4 Department of Comparative, Diagnostic, and Population Medicine, University of Florida, College of Veterinary Medicine, Gainesville, Florida, United States of America

\* erum.khan@aku.edu

# Abstract

# Background

Arboviruses are a cause of acute febrile illness and outbreaks worldwide. Recent outbreaks of Chikungunya virus (CHIKV) in dengue endemic areas have alarmed clinicians as unique clinical features differentiating CHIKV from Dengue virus (DENV) are limited. This has complicated diagnostic efforts especially in resource limited countries where lab testing is not easily available. Therefore, it is essential to analyse and compare clinical features of laboratory confirmed cases to assist clinicians in suspecting possible CHIKV infection at time of clinical presentation.

# Methodology

A prospective point prevalence study was conducted, with the hypothesis that not all patients presenting with clinical suspicion of dengue infections at local hospitals are suffering from dengue and that other arboviruses such as Chikungunya, West Nile viruses, Japanese Encephalitis virus and Zika virus are co-circulating in the Sindh region of Pakistan. Out-patients and hospitalized (in-patients) of selected district hospitals in different parts of Sindh province of Pakistan were recruited. Patients with presumptive dengue like illness (Syndromic diagnosis) by the treating physicians were enrolled between 2015 and 2017.

Current study is a subset of larger study mentioned above. Here-in we compared laboratory confirmed cases of CHIKV and DENV to assess clinical features and laboratory findings that may help differentiate CHIKV from DENV infection at the time of clinical presentation. analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Results

Ninety-eight (n = 98) cases tested positive for CHIKV, by IgM and PCR and these were selected for comparative analysis with DENV confirmed cases (n = 171). On multivariable analysis, presence of musculoskeletal [OR = 2.5 (95% CI:1.6–4.0)] and neurological symptoms [OR = 4.4 (95% CI:1.9–10.2)], and thrombocytosis [OR = 2.2 (95% CI:1.1–4.0)] were associated with CHIKV infection, while atypical lymphocytes [OR = 8.3 (95% CI:4.2-16.7)] and thrombocytopenia [OR = 8.1 (95% CI:1.7-38.8)] were associated with DENV cases at time of presentation. These findings may help clinicians in differentiating CHIKV from DENV infection.

# Conclusion

CHIKV is an important cause of illness amongst patients presenting with acute febrile illness in Sindh region of Pakistan. Arthralgia and encephalitis at time of presentation among patients with dengue-like illness should prompt suspicion of CHIKV infection, and laboratory confirmation must be sought.

# Author summary

Chikungunya and Dengue viral infection are prevalent in South Asia. These are spread by mosquito bites and present as acute febrile illness with clinical presentation difficult to differentiate. In this study we analysed clinical findings of patients with lab confirmed Dengue virus and Chikungunya virus infection and compared the clinical symptoms to assess significant features that can differentiate the two infections at time of presentation to clinics and hospitals. The study was conducted at the Aga Khan University Hospital, Karachi Pakistan. Our study confirms presence of Chikungunya virus as an etiologic agent of fever among patients with dengue like illness in Sindh province of Pakistan. Presence of joint pains, central nervous symptoms were more common in CHKV patients, while gastrointestinal symptoms and elevated liver enzymes were more commonly associated with dengue patients. Both viruses are endemic in Pakistan and clinical and lab diagnosis is essential.

# Introduction

Arbovirus infections are a disease spectrum associated with significant morbidity in humans. Amongst these, Dengue virus (DENV) and Chikungunya virus (CHIKV) share geographic and vector related features and mainly affect the tropics [1]. These vector-borne illnesses spread by the mosquitos of the *Aedes* family. While DENV belongs to the genus *Flavivirus*, CHIKV is a member of the genus *Alphavirus* in the family *Togaviridae*.

Both viral infections share similar disease manifestations including fever, rash and other non-specific findings. CHIKV presents after an incubation period of 2–3 days, as an abrupt onset of febrile illness with arthralgia and rash that resolves spontaneously in 7–10 days. However, in some patients, disease may progress to recurrent, debilitating polyarthralgia and polyarthritis lasting for months to years [2]. Additionally, some patients may present with signs of central nervous system involvement, making it essential to diagnose.

First recorded in Tanzania during 1952–53, CHIKV has caused multiple outbreaks worldwide involving all continents but Antarctica [3]. CHIKV has been circulating actively in the Indian subcontinent for the last decade with reports of epidemic outbreaks from India to Sri Lanka. India reported around 1.4 million CHIKV infections in 2006 [4].

Despite this active transmission in the region, reports of CHIKV activity have been reported from Pakistan only very recently [3–5] with the first outbreak reported from Karachi in 2017 [6–9]. The true burden of this virus in the community is largely unknown especially in the backdrop of reports that 20–30% of clinically diagnosed DENV cases being negative for the virus; referred to in this study as non-DENV acute febrile illness (NDFI) [10–11]. Thus possibility of CHIKV infection misdiagnosed as dengue like illness.

In this study, we analysed subset of data of an active surveillance study for patients with acute febrile illness and/or suspected DENV infection in 5 cities of Sindh province in Pakistan. This study aimed firstly to ascertain the presence of CHIKV as etiologic agent of NDFI in Sindh province, and secondly, to compare clinical features of laboratory confirmed CHIKV cases with DENV infection to identify clinical features and laboratory findings that may help differentiate CHIKV from DENV infection at the time of clinical presentation.

# Methods

#### **Ethics statement**

This study was approved by Institutional Review Board of the Aga Khan University Hospital (reference number 3183-PAT-ERC-14). Informed consent was obtained from all adult patients enrolled. The process of sample collection and questionnaire was explained to the children and adolescent patients in simple words. If they agreed, informed consent was taken from their parent(s)/ guardian(s).

A prospective, cross-sectional study was carried out to assess role of arboviruses namely: Chikungunya, West Nile and Japanese encephalitis virus as cause of fever in NDFI cases in the Sindh region of Pakistan from 2015–2017. Surveillance for febrile illness was conducted at five major cities: Karachi, Hyderabad, Sukkhar, Larkana and Mirpurkhas. The combined population of these cities is over 25.3 million and consists of a mix of urban, peri-urban and rural settings. For this comparative study, a subgroup analysis was performed on patients who tested positive for DENV and CHIKV.

#### Patient selection and data collection

We recruited patients from 5 district hospitals one in each city. Patients presenting with acute febrile illness to out -patient department (OPD patients) and those admitted with dengue like illness (see details below) were recruited over a period of 3 years from 2015–2017. Appointed medical officers at each field site enrolled patients after taking written consent in accordance with the Declaration of Helsinki. In case of children less than 18 years, consent was sought from parents after the patients gave verbal assent to participation and sampling. Consent was taken in local language. Inclusion criteria was patients between 10–60 years of age with a fever >over 38oC (>100.4oF) for 3–10 days, with either of the following symptoms: headache, myalgia, arthralgia, vomiting, macular/ petechial rash, bleeding diathesis, altered mental status, or acute flaccid paralysis. Since the other causes of acute febrile illness had to be ruled out, i.e. typhoid by blood culture, malaria parasite by blood film and CBC to assist clinicians in patient management, in addition to serum collected for the viral studies. Consequently a large volume of blood sample (25 ml) was required, and this was the main reason for not including very young and very old patients. Thus, the exclusion criteria were age less than 10 years and >60 years, presence of chronic liver disease, haematological malignancies or chronic illnesses

requiring use of steroids (>20mg/day of prednisolone for > 2 weeks, or other immunosuppressant). Non-probability consecutive sampling was used. Samples were de-identified and stored at -80°C after separation of serum. DENV and CHIKV testing was performed using a commercial IgM Enzyme-Linked Immunosorbent Assay (ELISA) (CHIKjj Detect IgM, InBios International, Inc, Seattle, WA) and reverse-transcription polymerase chain reaction (RT-PCR). Laboratory results, along with epidemiological and clinical data were recorded in a predefined questionnaire.

#### Clinical and laboratory testing

Clinical diagnosis and patient management were performed according to the hospital's standard of care. Details of the patient's clinical and laboratory profiles were entered in the predefined data collection form by a trained research medical officer. Clinical diagnosis was defined as the provisional diagnosis made by the treating physician at the time of presentation before the etiological diagnosis was established by the laboratory tests.

A laboratory diagnostic algorithm was followed (Fig 1). Samples that tested negative for DENV were considered as NDFI and were then evaluated for other viruses including CHIKV infection by IgM ELISA and RT-PCR. Hence, co-infection with dengue and chikungunya was not evaluated.

**Serologic interpretation- ELISA.** CHIKV and DENV immunoglobulin M (IgM) antibodies were detected via commercial IgM ELISA platforms (CHIKjj Detect IgM, InBios International, Inc, Seattle, WA), (DENV Detect IgM, InBios International, Inc, Seattle, WA) according to manufacturer instructions. These are sandwich-type immunoassays that detect human antibodies targeting the CHIKV and DENV envelope glycoproteins. Any indeterminate or borderline samples were retested and evaluated in duplicate to verify immune status. Dengue IgM was tested for cross-reactivity with other flaviviruses like West Nile, Japanese Encephalitis and Zika by Plaque Reduction Neutralization Test (PRNT). Dengue subtyping was done by PCR for antigen positive samples and PRNT for IgM positive samples.

**RT-PCR.** RNA extractions were performed using the QIAamp Viral RNA Mini kit (QIA-GEN, Hilden, Germany) and RT-PCR performed as described elsewhere [8, 10]. A cycle threshold (Ct) value of <40 cycles was regarded as positive.

#### Statistical analysis

Data from proforma was double entered into EpiData database. Data was then extracted in Microsoft Excel and transferred to SPSS 21.0 software for statistical analysis.

Frequencies and proportions were determined for qualitative variables, whereas, mean and standard deviation was calculated for quantitative variables. Pearson's Chi-Square test and Independent t-test was used to identify associations between variables.

#### Results

991 patients with a clinical diagnosis of DENV infection were enrolled. Of these 17.25% (n = 171) tested positive and confirmed as DENV infection. 11.95% (n = 98) of NDFI patients tested positive for CHIKV via IgM ELISA and RT-PCR. 79.6% (n = 78) cases of CHIKV were detected by IgM while 11.2% (n = 11/98) of samples tested positive for CHIKV by either ELISA or RT-PCR; fulfilling the CDC diagnostic criteria [12] for a confirmed infection. For Dengue fever, 63.1% (n = 108) of NS1Ag positive were also positive for IgM antibodies, while DENV was detected by PCR in 45.6% (n = 78) cases.

### Patient demographics

Patients with DENV were significantly younger than patients with CHIKV infection. DENV patients had a mean age of 33.6 years while CHIKV-infected patients had a mean age of 40.3 years (Table 1). More than 80% of patients in both groups were adults, with males comprising 56.7% and 55.10% of the population for DENV and CHIKV, respectively.



https://doi.org/10.1371/journal.pntd.0008086.g001

	CHIKV	DENV	
Continuous variable	Mean ±SD	Mean ±SD	Difference (95%CI)
Mean age [years]	40.2 ±17.1	33.6 ±15.1	6.6 (2.5 to 10.7)
Categorical variables	n = 98 (%)	n = 171 (%)	OR (95%CI)
Children	10 (10.2)	22 (12.9)	0.8 (0.3–1.7)
Adults	88 (89.8)	149 (87.1)	1.2 (0.6–1.9)
Male	54 (55.1)	97 (56.7)	1.0 (0.8–1.2)
Female	44 (44.9)	74 (43.2)	1.1 (0.8–1.4)
Location (total recruited)			
Karachi	89 (90.8)	149 (87.1)	-
Mirpurkhas	1 (1.0)	7 (4.1)	-
Sukkhar	2(2.0)	2 (1.2)	-
Larkana	1(1.0)	2 (1.2)	-
Hyderabad	5 (5.1)	11(6.4)	-

Table 1. Demographics of	patients diagnosed with either DENV or CHIKV. Bold font indicates stat	tistical significance

Demographics of patients with dengue and Chikungunya virus infections. Qualitative variable is represented with frequency count and proportion [n = (%)] whereas for quantitative variables, mean and standard deviation [mean (SD)] is given. OR = Odds ratio, 95% CI = 95% confidence interval.

https://doi.org/10.1371/journal.pntd.0008086.t001

The majority of patients with DENV were from Karachi (87.1%, n = 149), followed by Hyderabad (6.4%, n = 11) (Table 1). Karachi was also the site for most of the CHIKV positive cases (91%, n = 89), then Hyderabad (5.05%, n = 5) (Table 1). Upper Sindh showed less activity for both DENV and CHIKV infection with only 4% of DENV patients from Mirpurkhas and 1% from each Sukkhar and Larkana. For CHIKV infection, 2% of patients were enrolled from Sukkhar and 1% each from Mirpurkhas and Larkana.

#### **Clinical findings**

**Symptoms.** Most patients presented with a fever of less than 7 days with mean duration of 4.2 days in patients with DENV infection as compared to 5.4 days in patients with CHIKV infection.

Average fever of patients with CHIKV was 100.5°F and patients with DENV had an average temperature of 100.3°F. While there were differences in average systolic blood pressure, respirations and pulse, these variables were not statistically significant.

While Glasgow Coma Scale (GCS) did not differ significantly between the 2 groups, patients with confirmed CHIKV infection frequently presented with severe musculoskeletal pain and central nervous system (CNS) related symptoms (such as altered mentation, confusion and loss of consciousness): this difference was statistically significant with a p-value of 0.001. Gastrointestinal (GI) symptoms including nausea, abdominal pain and vomiting were reported in 53.8% of DENV patients (p<0.001) (Table 2). Haemorrhage was significantly associated with DENV virus infection with 8% of patients presenting with this symptom (p = 0.040) (Table 2). Rash was seen in 15.3% and 15.2% of CHIKV and DENV cases respectively and was not a significant predictor of disease (Table 2).

#### Laboratory tests

**Haematological tests.** Average haematocrit and haemoglobin levels were within normal range for both DENV and CHIKV patients though 34.1% of DENV patients exhibited elevated haematocrit p <0.001 (as compared to 20% of CHIKV patients). For DENV, thrombocytopenia (defined as platelet count <150,000/mm3). was a significant predictor of disease with

	CHIKV	DENV	Difference (95%CI)	OR CHIKV (95%CI)	OR DENV (95%CI)
Clinical presentation	Mean ±SD	Mean ±SD			
Mean days of fever	5.4±5.4	4.2±1.9	1.2 (0.1 to 2.4)	-	-
Mean temperature°C	100.5±2.0	100.2±1.9	0.2(-0.3 to 0.8)	-	-
Pulse (beats per minute)	97.2±16.2	95±19.7	2.3(-3.0 to 7.6)	-	-
Respirations (breaths per minute)	21.2±4.3	21±6.9	0.1(-1.6 to 1.9)	-	-
Mean systolic pressure	121±23.0	115.7±15.4	5.3 (-0.7 to 11.3)	-	-
Mean GCS	14.2±2.4	14.9±0.9	-0.7 (-1.2 to -0.2)	-	-
Symptoms	n (%)	n (%)			
Gastrointestinal	26 (26.5)	92 (53.8)	-	0.5 (0.3 to 0.7)	2.0(1.4 to 3.0)
Respiratory	2 (2.0)	2 (1.2)	-	1.7 (0.2 to 12.1)	0.6 (0.1 to 4.0)
Neurological	21(21.4)	12 (7.0)	-	3.0 (1.6 to 5.9)	0.3(0.2 to 0.6)
Haemorrhage	2 (2.0)	14 (8.1)	-	0.2 (0.1 to 1.1)	4.0 (0.9 to 17.3)
Musculoskeletal	61 (62.2)	51 (29.8)	-	2.1 (1.6 to 2.7)	0.5(0.4 to 0.6)
Ocular	9 (9.2)	11 (6.4)	-	1.4 (0.6 to 3.3)	0.7 (0.3 to 1.6)
Rash	15 (15.3)	26(15.2)	-	1.0 (0.6 to 1.8)	1.0 (0.5 to 1.8)
Laboratory parameters	Mean ±SD	Mean ±SD			
Aspartate aminotransferase (IU/L)	60.9±30.1	158.8±137.1	-97.9 (-125.5 to-67.2)	-	-
Alanine aminotransferase (IU/L)	55.1±53.3	113.4±115	-58.2 (-85.4 to-31.2)	-	-
Haemoglobin (g/dl)	12.1±2.5	13.6±2.0	-1.5 (-2.1 to -0.9)	-	-
Haematocrit (%)	37.2±6.6	40.5±5.6	-3.4 (-5.0to-1.7	-	-
Total leucocyte count (x10 <sup>9</sup> /l)	8.6±4.7	4.2±2.3	4.5 (3.4 to 5.4)	-	-
Platelets (x10 <sup>9</sup> /l)	254.7±150.1	110.8±89.8	143.5 (109.8 to 177.1)	-	-
Altered red blood cell morphology seen	36 (36.7)	35 (20.5)	-	1.8 (1.2 to 2.7)	0.6 (0.4 to 0.8)
Atypical lymphocytes seen	28 (28.6)	149(87.1)	-	0.3 (0.2 to 0.5)	3.0 (2.2 to 4.2)

Table 2. Clinical presentation and laboratory findings of acute febrile patients with Dengue virus and Chikungunya virus infection from Sindh, Pakistan 2015–2017 (Bold font indicates statistical significance).

Comparison of clinical presentation and laboratory profile amongst patients with dengue and Chikungunya virus infections. Qualitative variable is represented with frequency count and proportion [n = (%)] whereas for quantitative variables, mean and standard deviation [mean (SD)] is given. GCS = Glasgow Coma Scale, OR = Odds ratio, 95%CI = 95% confidence interval.

https://doi.org/10.1371/journal.pntd.0008086.t002

74.1% of patients exhibiting this phenomenon (p < 0.001) as compared to 18% of CHIKV patients; almost 150 platelets less than CHIKV (Table 2). While both DENV and CHIKV infected patients exhibited, on average, elevated leucocyte counts, 87% of DENV infected patients presented with atypical lymphocytes and were 3 times more likely to present with this sign than patients infected with CHIKV (Table 2). Altered red blood cell (RBC) morphology was also a significant finding and was reported almost twice as frequently in CHIKV infected patients than DENV infected patients, with 36.7% of CHIKV patients showing elevate levels of misshaped RBCs compared to only 20% of DENV patients (Table 2).

**Liver function tests.** Upon enrolment, the mean alanine aminotransferase (ALT) and mean aspartate aminotransferase (AST) levels of patients with CHIKV were within the normal range (55.13 IU/L and 60.90 IU/L, respectively) while DENV patients had significantly higher mean levels of ALT (113.4 IU/L) and AST (158.7 IU/L) (Table 2).

Patients with DENV were more likely to have elevated liver enzymes, 88.4% AST (OR 3.6, 95% CI: 1.2–10.7) and 58.3% ALT (OR 2.6, 95% CI: 1.2–5.6) than patients with CHIKV (68.1% AST and 35.1% ALT).

**Multivariable analysis.** On multivariable analysis (not shown in tables), presence of musculoskeletal [OR = 2.5 (95% CI:1.6–4.0)] and neurological symptoms [OR = 4.4 (95% CI:1.9–

· ·				
Provisional diagnosis	Chikungunya n = 98(%)	Dengue n = 171(%)	OR for CHIKV (95%CI)	OR for DENV (95%CI)
Viral infection	49 (50)	142 (83.0)	0.6 (0.5 to 0.7)	1.7 (1.3 to 2.0)
Bacterial infection	10 (10.2)	12 (7.0)	1.5 (0.7 to 3.2)	0.7 (.3 to 1.5)
Malaria	6 (6.1)	17 (9.9)	0.6 (0.3 to 1.5)	1.6 (0.7 to 4.0)
Suspected non-infectious aetiology	12 (12.2)	12 (7.0)	1.7 (0.8 to 3.7)	0.6 (0.3 to 1.2)
Therapy				
Antiviral agent	10 (1.02)	4 (2.3)	4.3 (1.4 to 13.4)	0.2 (0.1 to 0.7)
Systemic steroids	5 (5.1)	6(3.5)	1.5 (0.5 to 4.7)	0.7 (0.2 to 2.2)
Fluid replacement	50 (51.0)	131 (76.6)	0.6 (0.5 to 0.8)	1.5 (1.2 to 1.8)
Transfusions	1 (1.0)	4 (2.3)	0.4 (0.1 to 3.8)	2.3 (0.3 to 20.2)
Anti-malarial	1 (1.0)	3 (1.7)	0.6 (0.1 to 5.5)	1.7 (0.2 to 16.3)
Analgesics	59 (60.2)	106 (61.9)	1.0 (0.8 to 1.2)	1.0 (0.8 to 1.3)
Alive	92/92 (100)	170/170	-	-
			Difference (95%CI)	
Mean length of hospital stay (days)	3.1±5.2	3.5±2.6	-0.3 (-1.5 to 0.8)	

Table 3. Clinical diagnosis and management profile with outcomes of patients suffering from acute Chikungunya and Dengue enrolled from Sindh, Pakistan [2015-2017].

Clinical diagnosis and management profile amongst patients with dengue and Chikungunya virus infections. Qualitative variable is represented with frequency count and proportion [n = (%)] whereas for quantitative variables, mean and standard deviation [mean (SD)] is given. OR = Odds ratio, 95%CI = 95% confidence interval.

https://doi.org/10.1371/journal.pntd.0008086.t003

10.2)], and thrombocytosis [OR = 2.2 (95% CI:1.1-4.0)] were associated with CHIKV infection, while atypical lymphocytes [OR = 8.3 (95% CI:4.2-16.7)] and thrombocytopenia [OR = 8.1 (95% CI:1.7-38.8)] were associated with DENV cases at the time of presentation.

**Clinical management.** At presentation, 83.0% of DENV patients were clinically diagnosed with a viral infection, while only 50% of patients in the CHIKV group were diagnosed with a viral infection (Table 3). 10.2% of CHIKV patients were diagnosed with a bacterial infection and 12% as having a non-infectious aetiology by physicians (Table 3). Malaria was suspected in 10% of DENV infected patients and 6% of CHIKV patients (Table 3).

Despite have fewer viral diagnoses, CHIKV infected patients were 4 times more likely to receive antiviral therapy (primarily acyclovir) than DENV patients (Table 3). While not significant, use of steroid therapy was documented in 5.1% of CHIKV patients as compared to 3.5% of patients with DENV (Table 3). Fluid replacement therapy was required in 76.6% of DENV patients and 51% of CHIKV infected patients (Table 3). Up to 60% of patients required analgesics in both groups (Table 3).

#### Discussion

In this study, we show that CHIKV is an important cause of NDFI cases in the Sindh province (12.1%) cases. These values are in contrast to those reported from other countries in the Asian region. Reller et al from Sri Lanka, in 2015 reported acute CHIKV infection in 3.5% of patients with acute febrile illness, 82.7% remained undiagnosed after complete laboratory workup [10]. Similarly, studies from India have reported CHIKV prevalence of 9.6–25% while 37% remained undiagnosed [6,11]. In our study most cases were from Karachi; one of the most densely populated cities in the world with a variety of social/economic factors contributing to uncontrolled urbanization and poor sanitation standards. This study also included samples from cluster outbreak of CHIKV infection during November 2016 that made the headline news in the country with 30,000 patients suspected of CHIKV infection [9].

Co-circulation of DENV and CHIKV in Karachi reflects a progressive deterioration of municipal infra-structure and efforts to control mosquitos [13]. More concerning is the identification of DENV and CHIKV in less densely populated peri-urban and rural areas, which lack of basic public health facilities, piped water and municipal infrastructure, making future epidemics imminent as seen in neighbouring India [14], unless preventive strategies are implemented in a timely manner.

Clinically, longer duration of fever, severe arthralgia and neurological manifestations were significantly associated with CHIKV. Twenty-one percent of CHIKV positive patients presented with CNS manifestations. This may have prompted the use of acyclovir in management of patients with CHIKV infection at time of admission, in lieu of possible herpes encephalitis. Similar findings of CNS involvement at time of presentation have been reported from India where altered mental status was seen in 56.67% of CHIKV patients requiring admission to intensive care units [15]. Literature from most of the CHIKV endemic region (i.e. Indian Ocean, South Asia, the Pacific islands, Southern Europe, the Caribbean, and South America) have reported neurological manifestation of CHIKV [16].

A study in La Reunion reported 24.1% of CHIKV cases presenting with abnormal neurological signs and symptoms [17]. And a recent study from Pakistan reported neurological symptoms in almost half of patients with confirmed CHIKV infections [8].

GI manifestations were most pronounced in patients with DENV infection, with 53.8% of DENV cases as compared to 26.2% of CHIKV, a finding that may help differentiate the two infections in clinical setting. The association of DENV with GI symptoms has consistently been reported from other endemic regions. Ramos et al in their retrospective review of 8559 patients reported GI symptoms in 67% of DENV patients [18]. In vitro studies have also shown DENV to have a predilection for GI tissue, especially liver [19]. This may explain the elevated AST and ALT levels in the DENV group and other reports from endemic regions [20]. Fernando et al recommend testing for liver function enzymes after 4–5 days of illness, as low levels of liver enzymes at the early stage of disease may underestimate ongoing liver damage [21]. Similarly, a study from Bangkok estimated up to a ten-fold increase of ALT/AST from baseline in hepatitis associated with severe DENV cases [22].

These findings are significant for Pakistan as chronic hepatitis B and C along with other gastrointestinal infections are endemic in the population. This data should alert physicians to include DENV in the differential diagnosis of patients presenting with fever, hepatitis and acute gastrointestinal symptoms.

Pronounced musculoskeletal symptoms including polyarthralgia was reported in 62.2% of CHIKV patients as compared to 29.8% in DENV patients. These symptoms may have resulted in misdiagnosis of CHIKV infection as non-infectious arthritis by clinicians at time of presentation and resultant use of steroids in these patients.

No significant difference was seen in the frequency of rash in both groups. The presence of comparatively lower systolic pressures in DEN-V and increased levels of haematocrit likely reflect plasma leakage resulting from increased vascular permeability and third spacing. [18]. Fluid replacement and supportive management was required more frequently in DEN-V patients, a finding in concordance with other reports from Pakistan [23, 24] and other parts of dengue endemic region [1,25].

Higher leucocyte counts and atypical lymphocytes were observed for DENV. Significant thrombocytopenia was also found in DENV patients. This corroborates the findings of Lee et al in their study on CHIKV versus DENV infections in adults [1]. While case fatality has been associated with both DENV and CHIKV, all patients in our study survived, [15, 26–29].

For laboratory diagnostics, the Dengue Early Rapid NS-1 test was the most effective for DENV detection as most cases of DENV infection were detected using this assay. Many of

these patients were also positive by RT-PCR. Most NS-1 negative cases in our study were also negative by RT-PCR, validating the clinical utility of the NS-1 assay. While factors such as specimen shipping and delayed preservation could be possible reasons for the false negative RT-PCR cases in our study population, our findings agree with Ahmed et al who found that the overall performance of DENV RT-PCR to be no better than NS-1 antigen detection [30].

For the CHIKV serological diagnosis, 11% were positive with both IgM ELISA and RT-PCR, while 31.3% were detected only by RT-PCR, indicating that use of IgM alone in clinical settings as diagnostic assay may be insufficient and may result in false negative cases. This may happen in early phase of disease when antibodies are not detectable due to neutralization with viral antigen.

Nucleotide sequencing reports have shown associations of specific mutations at residue 226 of membrane fusion glycoprotein E1 (E1-A226V) region with disease severity and better virus adaptability [31]. These mutations have been associated with abrupt epidemic outbreaks of CHIKV. A study from La Reunion reported 34.3% of the total island population infected with viral containing the E1-A226V mutation [31]. Whole genome sequencing studies are currently underway to identify any mutations in CHIKV isolates yielded from Pakistan.

Most of our CHIKV patients were adult males which is in concordance with CHIKV epidemic reports from India and other regions [32, 33]. One sero-survey conducted in Calcutta reported the highest infection rate adults aged 50–55 years and virtually no sero-positivity in young children [34]. Further studies are required to explore this apparent age predilection in our population especially in very young children as this group was not included in this study.

Our study is limited by the nested design on a cross-sectional surveillance of febrile patients who presented to health centres. Since there was no follow-up, the outcomes and long-term sequelae of CHIKV infections could not be assessed. Disease severity data was also not collected; hence the subgroup of severe dengue cases could not be compared with CHIKV cases admitted to the hospital. Another shortcoming was the testing algorithm: as only DENV NS1 negative cases were tested for CHIKV, co-infections and specific manifestations in such cases could not be determined.

# Conclusion

Our study confirms that CHIKV is present in Sindh region of Pakistan and is an important cause of acute febrile illness. Patients with CHIKV infection are more likely to present with severe arthralgia and neurological symptoms. The recent CHIKV outbreak in Karachi high-lights the possibility of Karachi having an epidemic potential in future, unless timely preventive interventions are ensured. While the sample size and the study design was insufficient to calculate the actual prevalence of disease in the community, our findings raise concerns for the possible circulation of a virulent CHIKV genotype known for its high rate of CNS involvement.

# **Supporting information**

**S1 Checklist. STROBE checklist.** (DOC)

**S1 Table.** Primer and probe sequences used for Chikungunya and dengue virus detection. (DOCX)

# **Author Contributions**

Conceptualization: Joveria Q. Farooqi, Kelli L. Barr, Erum Khan.

Data curation: Faisal R. Malik.

Formal analysis: Uzma Shahid.

Funding acquisition: Maureen T. Long, Erum Khan.

Investigation: Uzma Shahid.

- Methodology: Uzma Shahid, Joveria Q. Farooqi, Kelli L. Barr, Kehkashan Imitaz, Zahida Azizullah, Dhani Prakoso, Maureen T. Long, Erum Khan.
- Project administration: Joveria Q. Farooqi, Kelli L. Barr, Faisal R. Malik, Maureen T. Long, Erum Khan.

Software: Faisal R. Malik.

Supervision: Faisal R. Malik, Maureen T. Long, Erum Khan.

Validation: Kehkashan Imitaz, Zahida Azizullah, Dhani Prakoso.

Writing - original draft: Uzma Shahid.

Writing – review & editing: Uzma Shahid, Joveria Q. Farooqi, Kelli L. Barr, S. Faisal Mahmood, Bushra Jamil, Kehkashan Imitaz, Zahida Azizullah, Faisal R. Malik, Dhani Prakoso, Maureen T. Long, Erum Khan.

#### References

- Lee VJ, Chow A, Zheng X, Carrasco LR, Cook AR, Lye DC, et al. Simple clinical and laboratory predictors of Chikungunya versus dengue infections in adults. PLoS neglected tropical diseases. 2012; 6(9): e1786. Epub 2012/10/03. https://doi.org/10.1371/journal.pntd.0001786 PMID: 23029573
- Elsinga J, Gerstenbluth I, van der Ploeg S, Halabi Y, Lourents NT, Burgerhof JG, van der Veen HT, Bailey A, Grobusch MP, Tami A. Long-term Chikungunya Sequelae in Curaçao: Burden, Determinants, and a Novel Classification Tool, J Infect Dis 2017; 216(5):573–581, https://doi.org/10.1093/infdis/jix312 PMID: 28931219
- Pialoux G, Gauzere BA, Jaureguiberry S, Strobel M. Chikungunya, an epidemic arbovirosis. The Lancet Infectious diseases. 2007; 7(5):319–27. Epub 2007/04/24. https://doi.org/10.1016/S1473-3099(07) 70107-X PMID: 17448935
- 4. Ravi V. Re-emergence of chikungunya virus in India. Indian journal of medical microbiology. 2006; 24 (2):83–4. Epub 2006/05/12. https://doi.org/10.4103/0255-0857.25175 PMID: 16687855
- Ray P, Ratagiri VH, Kabra SK, Lodha R, Sharma S, Sharma BS, et al. Chikungunya infection in India: results of a prospective hospital based multi-centric study. PloS one. 2012; 7(2):e30025. Epub 2012/03/ 01. https://doi.org/10.1371/journal.pone.0030025 PMID: 22363413
- 6. Rauf M, Fatima Tuz Z, Manzoor S, Mehmood A, Bhatti S. Outbreak of chikungunya in Pakistan. The Lancet Infectious diseases. 2017; 17(3):258. Epub 2017/03/01.
- Aamir UB, Badar N, Salman M, Ahmed M, Alam MM. Outbreaks of chikungunya in Pakistan. The Lancet Infectious diseases. 2017; 17(5):483. Epub 2017/04/28. https://doi.org/10.1016/S1473-3099(17) 30191-3 PMID: 28447956
- Barr KL, Khan E, Farooqi JQ, Imtiaz K, Prakoso D, Malik F, et al. Evidence of chikungunya virus disease in Pakistan since 2015 with patients demonstrating involvement of the central nervous system. Frontiers in public health. 2018; 6:186. https://doi.org/10.3389/fpubh.2018.00186 PMID: 30042937
- News T. Dengue-like undiagnosed viral fevers on the rise in Pakistan'. The News; 2015 [updated March 25, 2015; cited 2018 3 June, 2018]; Available from: https://www.thenews.com.pk/print/31136-dengue-like-undiagnosed-viral-fevers-on-the-rise-in-pakistan.
- Khan E, Farooqi JQ, Barr KL, Prakoso D, Nasir A, Kanji A, et al. Flaviviruses as a Cause of Undifferentiated Fever in Sindh Province, Pakistan: A Preliminary Report. Frontiers in public health. 2016; 4:8. Epub 2016/02/26. https://doi.org/10.3389/fpubh.2016.00008 PMID: 26909342
- CTaL JM. (Ed.) EpiData—Comprehensive Data Management and Basic Statistical Analysis System. Odense Denmark, EpiData Association, 2010. 2010.

- Centers. for. Disease. Control and prevention. Arboviral Diseases, Neuroinvasive and Non-neuroinvasive | 2015 Case Definition. Atlanta: Department of Health & Human Services; (2017).
- Naqvi S, Bashir S, Rupareliya C, Shams A, Giyanwani PR, Ali Z, et al. Clinical Spectrum of Chikungunya in Pakistan. Cureus. 2017; 9(7):e1430. Epub 2017/09/20. https://doi.org/10.7759/cureus.1430 PMID: 28924518
- Chahar HS, Bharaj P, Dar L, Guleria R, Kabra SK, Broor S. Co-infections with chikungunya virus and dengue virus in Delhi, India. Emerging infectious diseases. 2009; 15(7):1077–80. Epub 2009/07/25. https://doi.org/10.3201/eid1507.080638 PMID: 19624923
- Gupta A, Juneja D, Singh O, Garg SK, Arora V, Deepak D. Clinical Profile, Intensive Care Unit Course, and Outcome of Patients Admitted in Intensive Care Unit with Chikungunya. Indian journal of critical care medicine: peer-reviewed, official publication of Indian Society of Critical Care Medicine. 2018; 22 (1):5–9. Epub 2018/02/10.
- Mehta R, Gerardin P, de Brito CAA, Soares CN, Ferreira MLB, Solomon T. The neurological complications of chikungunya virus: A systematic review. Reviews in medical virology. 2018; 28(3):e1978. Epub 2018/04/20. https://doi.org/10.1002/rmv.1978 PMID: 29671914
- Economopoulou A, Dominguez M, Helynck B, Sissoko D, Wichmann O, Quenel P, et al. Atypical Chikungunya virus infections: clinical manifestations, mortality and risk factors for severe disease during the 2005–2006 outbreak on Reunion. Epidemiology and infection. 2009; 137(4):534–41. Epub 2008/08/ 13. https://doi.org/10.1017/S0950268808001167 PMID: 18694529
- Ramos-De La Medina A, Remes-Troche JM, Gonzalez-Medina MF, Anitua-Valdovinos Mdel M, Ceron T, Zamudio C, et al. [Abdominal and gastrointestinal symptoms of Dengue fever. Analysis of a cohort of 8559 patients]. Gastroenterologia y hepatologia. 2011; 34(4):243–7. Epub 2011/04/09. Sintomas abdominales y gastrointestinales del dengue. Analisis de una cohorte de 8.559 pacientes. <u>https://doi.org/10. 1016/j.gastrohep.2011.01.012</u> PMID: 21474206
- Martina BE, Koraka P, Osterhaus AD. Dengue virus pathogenesis: an integrated view. Clinical microbiology reviews. 2009; 22(4):564–81. Epub 2009/10/14. <u>https://doi.org/10.1128/CMR.00035-09</u> PMID: 19822889
- Iqtadar S, Akbar N, Huma N, Randhawa FA. Profile of hepatic involvement in dengue infections in adult Pakistani population. Pakistan journal of medical sciences. 2017; 33(4):963–7. Epub 2017/10/27. https://doi.org/10.12669/pjms.334.13026 PMID: 29067074
- Fernando S, Wijewickrama A, Gomes L, Punchihewa CT, Madusanka SD, Dissanayake H, et al. Patterns and causes of liver involvement in acute dengue infection. BMC infectious diseases. 2016; 16:319. Epub 2016/07/09. https://doi.org/10.1186/s12879-016-1656-2 PMID: 27391896
- Martinez Vega R, Phumratanaprapin W, Phonrat B, Dhitavat J, Sutherat M, Choovichian V. Differences in Liver Impairment Between Adults and Children with Dengue Infection. The American journal of tropical medicine and hygiene. 2016; 94(5):1073–9. Epub 2016/03/16. <u>https://doi.org/10.4269/ajtmh.15-</u> 0507 PMID: 26976884
- Jahan F. Dengue fever (DF) in Pakistan. Asia Pacific family medicine. 2011; 10(1):1. <u>https://doi.org/10.1186/1447-056X-10-1 PMID: 21349169</u>
- 24. Raza FA, ur Rehman S, Khalid R, Ahmad J, Ashraf S, Iqbal M, et al. Demographic and clinico-epidemiological features of dengue fever in Faisalabad, Pakistan. PloSOne 2014; 9(3):e89868.
- Organization WH, Research SPf, Diseases TiT, Diseases WHODoCoNT, Epidemic WHO, Alert P. Dengue: guidelines for diagnosis, treatment, prevention and control: World Health Organization; 2009.
- Ali N, Usman M, Syed N, Khurshid M. Haemorrhagic manifestations and utility of haematological parameters in dengue fever: a tertiary care centre experience at Karachi. Scandinavian journal of infectious diseases. 2007; 39(11–12):1025–8. Epub 2007/09/14. https://doi.org/10.1080/ 00365540701411492 PMID: 17852892
- Pancharoen C, Thisyakorn U. Neurological manifestations in dengue patients. The Southeast Asian journal of tropical medicine and public health. 2001; 32(2):341–5. Epub 2001/09/15. PMID: <u>11556587</u>
- Riaz MM, Mumtaz K, Khan MS, Patel J, Tariq M, Hilal H, et al. Outbreak of dengue fever in Karachi 2006: a clinical perspective. JPMA The Journal of the Pakistan Medical Association. 2009; 59(6):339– 44. Epub 2009/06/19. PMID: 19534364
- Tournebize P, Charlin C, Lagrange M. [Neurological manifestations in Chikungunya: about 23 cases collected in Reunion Island]. Revue neurologique. 2009; 165(1):48–51. Epub 2008/10/07. Manifestations neurologiques du chikungunya: a propos de 23 cas colliges a la Reunion. <u>https://doi.org/10.1016/j.</u> neurol.2008.06.009 PMID: 18835614
- Ahmed NH, Broor S. Comparison of NS1 antigen detection ELISA, real time RT-PCR and virus isolation for rapid diagnosis of dengue infection in acute phase. Journal of vector borne diseases. 2014; 51 (3):194–9. Epub 2014/09/26. PMID: 25253212

- Tsetsarkin KA, Vanlandingham DL, McGee CE, Higgs S. A single mutation in chikungunya virus affects vector specificity and epidemic potential. PLoS pathogens. 2007; 3(12):e201. Epub 2007/12/12. <a href="https://doi.org/10.1371/journal.ppat.0030201">https://doi.org/10.1371/journal.ppat.0030201</a> PMID: 18069894
- Patil SS, Patil SR, Durgawale PM, Patil AG. A study of the outbreak of Chikungunya fever. Journal of clinical and diagnostic research: JCDR. 2013; 7(6):1059–62. Epub 2013/08/02. https://doi.org/10.7860/ JCDR/2013/5330.3061 PMID: 23905103
- Reller ME, Acarida U, Nagahawatte A, Devasiri V, Kodikaarachchi W, Strouse JJ, et al. Chikungunya as a cause of acute febrile illness in southern Sri Lanka. PloS one. 2013; 8(12): e82259. Epub 2013/12/ 07. https://doi.org/10.1371/journal.pone.0082259 PMID: 24312651
- Neogi DK, Bhattacharya N, Mukherjee KK, Chakraborty MS, Banerjee P, Mitra K, et al. Serosurvey of chikungunya antibody in Calcutta metropolis. The Journal of communicable diseases. 1995; 27(1):19– 22. Epub 1995/03/01. PMID: 7636147